
\documentclass[reqno]{amsart}
\usepackage{a4wide}
\usepackage{hyperref}

\AtBeginDocument{{\noindent\small
\emph{Electronic Journal of Differential Equations},
Vol. 2026 (2026), No. 39, pp. 1--13.\newline
ISSN: 1072-6691. URL: https://ejde.math.txstate.edu, DOI: 10.58997/ejde.2026.39}
\thanks{\copyright 2026. This work is licensed under a CC BY 4.0 license.}
\vspace{8mm}}

\begin{document}
\begin{thebibliography}{10}

\bibitem{jaadeo11} J. Abdullaev, A.S. Desyatnikov, E.A. Ostravoskaya;
Suppression of collapse for matter waves with orbital angular momentum, 
\emph{J. Optics}, \textbf{13} (2011), 064023.

\bibitem{AOR} R. P. Agarwal, D. O'Regan;
Boundary value problems for discrete equations, 
\emph{Appl. Math. Letters}, \textbf{10} (1997), 83--87.

\bibitem{AW} R. P. Agarwal, F. Wong;
Upper and lower solutions method for higher-order discrete boundary value problems, 
\emph{Math. Inequal. App}l., \textbf{1} (1998) 551--557.


\bibitem{pacr15} P. Amster, C. Rogers;
On a Ermakov-Painlev\'e II reduction in three-ion electrodiffusion. 
A Dirichlet boundary value problem,
\emph{Discrete and Continuous Dynamical Systems}, Series A, \textbf{35} (2015), 3277--3292.


\bibitem{ha11} H. An;
Numerical pulsrodons of the (2+1)-dimensional rotating shallow water theory, 
\emph{Phys. Lett.} A, \textbf{375} (2011), 1921--1925.

%\bibitem{cacrurao90} C. Athorne, C. Rogers, U. Ramgulam and A. Osbaldestin,On linearisation of the Ermakov system,\emph{Phys. Lett.} A\textbf{143} (1990), 207--212.

\bibitem{lb64} L. Bass,
Irreversible interactions between metals and electrolytes,
\emph{Proc. Roy. Soc. London} A, \textbf{277} (1964), 125--136.

\bibitem{CO} A. Cabada, V. Otero-Espinar;
{Optimal existence results for $n$-th order periodic boundary value difference problems},
 \emph{J. Math. Anal. Appl}., \textbf{247} (2000), 67--86.

\bibitem{pc03} P. Clarkson,
Painlev\'e equations. Nonlinear special functions,
\emph{J. Comput. Appl. Math.}, \textbf{153} (2003), 127--140.

\bibitem{clark} P. Clarkson C. Dunning, B. Mitchell;
\emph{Discrete equations from B\"acklund transformations of the fifth Painlev\'e equation}. 
Preprint arxiv, available in https://arxiv.org/abs/2602.09756

\bibitem{rc99} R. Conte (Editor);
\emph{The Painlev\'e Property: One Century Later},
Springer, New York, 1999.

\bibitem{fokas} A. S. Fokas, B. Grammaticos, A. Ramani;
 From continuous to discrete Painlev\'e equations,
 J. Math. Anal. Appl, 180 (1993), 342--360.

\bibitem{fcmlbz90} F. Cornolti, M. Lucchesi, B. Zambon;
 Elliptical Gaussian beam self-focussing in nonlinear media, 
 \emph{Opt. Commun.}, \textbf{75} (1990), 129--135.


\bibitem{addbmdyk10} A. S. Desyatnikov, D. Buccoliero, M. R. Dennis, Y.S. Kivshar;
Suppression of collapse for spiralling elliptic solitons,
\emph{Phys. Rev. Lett.}, \textbf{104} (2010), 053902-1-4.

\bibitem{ve80} V. P. Ermakov;
Second order differential equations: conditions of complete integrability,
 Univ. Izy. Kiev, \textbf{20} (1880) 1--25.

\bibitem{FORP} D. Franco, D. O'Regan, J. Per\'an;
{Upper and lower solution theory for first and second order difference equations}. 
\emph{Dynamic Systems and Applications}, \textbf{13} (2004), 273--282.

\bibitem{kho}
A. Khovanskii;
{A class of systems of transcendental equations}, 
\emph{Dokl. Akad. Nauk}., \textbf{255} no. 4 (1980), 804--807.

\bibitem{agylas93} A. M. Goncharenko, Y. A. Logvin, A. M. Samson;
Self-focussing of two orthogonally polarized light beams in a nonlinear medium,
\emph{Opt. Quanturo Electron.}, \textbf{25} (1993), 97--104.

\bibitem{agylasps91} A. M. Goncharenko, Y. A. Logvin, A. M. Samson, P.S. Shapovalev;
Rotating ellipsoidal gaussian beams in nonlinear media,
\emph{Opt. Commun.}, \textbf{81} (1991), 225--230.

\bibitem{agylaspsst91} A. M. Goncharenko, Y. A. Logvin, A. M. Samson, P. S. Shapovalev,
S.I. Turovets;
Ermakov Hamiltonian systems in nonlinear optics of elliptic Gaussian Beams,
\emph{Phys. Lett.} A, \textbf{160} (1991), 138--142.


\bibitem{cgjmay72} C. R. Giuliano, J .H. Marburger, A. Yariv;
Enhancement of self-focussing threshold in sapphire with elliptical beams,
\emph{Appl. Phys. Lett.}, \textbf{21} (1972), 58--60.

\bibitem{jan} T. Jankowski;
On some difference problems, \emph{Bull. Austral. Math. Soc.}, 61 (1) (2000), 161--166.

\bibitem{luma} Y. Lu, R. Ma;
On positive periodic solutions of second-order difference equations with 
attractive-repulsive singularities, \emph{Adv. Difference Equ.}, 
2012 (2012), Article ID 186.

\bibitem{wn82} W. Nernst;
Zur kinetik der in L\"osung befindlichen k\"orper: Theorie der Diffusion,
\emph{Z. Phys. Chem.}, \textbf{2} (1882), 613--637.

\bibitem{mp90} M. Planck;
\"Uber die Erregung von Elektrict\"at und W\"arme in Electrolylen,
\emph{Ann. Phys. Chem.}, \textbf{39} (1890), 161--186.

\bibitem{irena} I Rachunkov\'a, L. Rachunek;
Singular discrete and continuous mixed boundary value problems,
\emph{Math. Comput. Modelling}, 49 (2009), 833--843.


\bibitem{cr89} C. Rogers,
Elliptic warm-core theory: the pulsrodon,
\emph{Phys. Lett.} A, \textbf{138} (1989), 267--273.

\bibitem{crha10} C. Rogers, H. An;
Ermakov-Ray-Reid systems in 2+1-dimensional rotating shallow water theory, 
\emph{Stud. Appl. Math.}, \textbf{125} (2010), 275--299.

\bibitem{crbmha12} C. Rogers, B. Malomed, H. An;
Ermakov-Ray-Reid reductions of variational approximations in nonlinear optics,
\emph{Stud. Appl. Math.}, \textbf{129} (2012), 389--413.

\bibitem{crbmkcha10} C. Rogers, B. Malomed, K. Chow, H. An;
Ermakov-Ray-Reid systems in nonlinear optics,
\emph{J. Phys. A: Math. Theor.}, \textbf{43} (2010), 455214 (15pp).

\bibitem{crws18} C. Rogers, W. K. Schief;
\emph{Ermakov-type systems in nonlinear physics and continuum mechanics, 
in Nonlinear Systems and Their Remarkable Mathematical Structures}, 
Editor Norbert Euler, CRC Press (2018) 541--576.

\bibitem{crws16} C. Rogers, W.K. Schief;
On Ermakov-Painlev\'e II systems. Integrable reduction,
\emph{Meccanica}, \textbf{51} (2016), 2967--2974.

\bibitem{crws11} C. Rogers, W. K. Schief;
The pulsrodon in 2+1-dimensional magneto-gasdynamics. Hamiltonian structure and 
integrability,
\emph{J. Math. Phys.}, \textbf{52} (2011), 083701 (20pp).

\bibitem{crws94} C. Rogers, W. K. Schief;
Multi-component Ermakov systems: structure and linearization,
\emph{J. Math. Anal. Appl.}, \textbf{198} (1994), 194--220.

\bibitem{crws97} C. Rogers, W.K. Schief;
The classical B\"acklund transformation and integrable discretisation of 
characteristic equations, \emph{Phys. Lett. A}, \textbf{232} (1997), 217--223.

\bibitem{wwhhjm68} W. G. Wagner, H. A. Haus, J. H. Marburger;
Large-scale self-trapping of optical beams in the paraxial ray approximation,
\emph{Phys. Res.}, \textbf{175} (1968), 256--266.

\bibitem{wang} J. Wang;
On systems of polynomials with at least one positive real zero,
\emph{Journal of Algebra and Its Applications}, \textbf{19} no. 10 (2020), 
art. 2050183.

\end{thebibliography}

\title[\hfilneg EJDE-2026/39\hfil Discretisation of the Ermakov-Painlev\'e II equation]
{Discretisation of the Ermakov-Painlev\'e II equation:
Dirichlet and Robin-type boundary value problems}


\author[P. Amster, C. Rogers \hfil EJDE-2026/39\hfilneg]
{Pablo Amster, Colin Rogers}

\address{Pablo Amster \newline
Departamento de Matem\'atica,
Facultad de Ciencias Exactas y Naturales,
Universidad de Buenos Aires, and
IMAS - CONICET, Ciudad Universitaria, Pabell\'on I,
(1428) Buenos Aires, Argentina}
\email{pamster@dm.uba.ar}

\address{Colin Rogers \newline
School of Mathematics and Statistics,
The University of New South Wales,
Sydney, NSW 2052, Australia}
\email{c.rogers@unsw.edu.au}

\thanks{Submitted December 3, 2025. Published June 8, 2026.}
\subjclass[2020]{39A27, 34B15}
\keywords{Ermakov-Painlev\'e II equation; Difference equations; Dirichlet boundary conditions;
\hfill\break\indent nonlinear Robin conditions}

\begin{abstract}
Two-point boundary value problems for a discrete Ermakov-Painle-v\'e II equation
are analysed by means of topological methods. In addition, an alternative variational
approach is detailed. Existence of solutions is established for
appropriate choice of parameters.
\end{abstract}

\maketitle
\numberwithin{equation}{section}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{remark}[theorem]{Remark}
\allowdisplaybreaks

\section{Introduction}

Ermakov-type nonlinear coupled systems with genesis in the classical work \cite{ve80}
have diverse physical applications in both physics and continuum mechanics \cite{crws18}.
These occur, \emph{inter alia} in nonlinear optics
\cite{fcmlbz90,agylas93, agylasps91, agylaspsst91, cgjmay72,crbmha12, crbmkcha10, wwhhjm68},
spiralling elliptic soliton analysis \cite{addbmdyk10}, Bose-Einstein condensate
theory \cite{jaadeo11}, oceanographic warm-core eddy evolution \cite{cr89},
2+1-dimensional magnetogasdynamics \cite{crws11} and rotating shallow water system
theory \cite{ha11, crha10}.
A key feature of these systems is the existence of a so-called
\emph{Ermakov invariant}, namely a conserved quantity that underpins their
integrability and allows for the derivation of nonlinear superposition principles.

On the other hand, the classical Painlev\'e equations PI-PVI are well-established
for their role in modern soliton theory and integrable systems 
\cite{pc03,rc99}.
In particular, the Painlev\'e II equation
notably has diverse physical applications not only in a solitonic context but also
in boundary value problems associated with the Nernst-Planck model of ion transport
\cite{wn82,mp90}.
In \cite{lb64}, Painlev\'e II was derived \emph{ab initio} in the analysis of a
boundary value problem which determines the electric field distribution in a region
occupied by an electrolyte.

More recently, a significant advancement has been the introduction of hybrid
\emph{Ermakov-Painlev\'e} systems, which merge the structure of Ermakov models with
the profound integrability properties of the Painlev\'e equations.
These hybrids have extended the applicability of Ermakov systems to new areas,
including cold plasma physics, electrodiffusion, and capillarity theory.
A distinguished member of the family of the hybrid {Ermakov-Painlev\'e} systems
is the EP-II equation, namely
\begin{equation}\label{eq-cont}
 y''(z)= Ay(z)^3 + Bzy(z) + \frac C{y(z)^3},\quad z\in (0,1).
\end{equation}
which provides a rich framework for studying nonlinear wave propagation and other
complex dynamics. Boundary value problems for this equation
have been recently studied in \cite{pacr15},
however, for numerical simulation and the analysis of discrete physical systems,
a discrete analogue is essential. This motivates the study of discretizations that
faithfully preserve the core mathematical properties of the original continuous system.


In this paper, we undertake a systematic study of a discrete version of the
Ermakov-Painlev\'e II equation. Specifically, we investigate the finite difference
equation
\begin{equation}\label{disc-eq}
\Delta^2 u_{x-1} = a u_{x}^3 + b x u_{x} + \frac{c}{u_x^3}, \quad x=1, \ldots, N-1,
\end{equation}
subject to two types of boundary conditions: the Dirichlet condition
\begin{equation}\label{dir}
u_0 = D_0, \quad u_N = D_N, \quad D_0, D_N > 0,
\end{equation}
and the nonlinear Robin condition
\begin{equation}\label{disc-robin}
\Delta u_0 = f_0(u_0), \quad \Delta u_{N-1} = f_N(u_N),
\end{equation}
where $f_0$, $f_N : (0, +\infty) \to \mathbb{R}$ are given continuous functions.
Our objective is to establish the existence of positive solutions for these discrete
boundary value problems.
It is worth observing that, in the literature on singular second-order equations,
the cases $c>0$ and $c<0$ are usually referred to as repulsive and attractive,
respectively, and their analysis often requires different mathematical tools.
For a discrete version of this terminology in difference equations, see \cite{luma}.

The literature on discretisations of Painlev\'e-type equations often focuses on the
preservation of structural features such as integrability or B\"acklund transformations.
Indeed, the derivation of discrete Painlev\'e equations from B\"acklund and Schlesinger
transforms of their continuous counterparts was introduced by Fokas, Grammaticos
and Ramani \cite{fokas}
and remains an active area of research (see, e.g., \cite{clark, crws97}).
However, in the present work, our primary goal is not the preservation of such
invariants but rather the analysis of boundary value problems for the discrete equation
using topological and variational methods. The standard finite difference scheme
adopted here, while not designed to preserve the Ermakov invariant, is sufficient for
the existence and uniqueness analysis we undertake. Moreover, the convergence of this
scheme to the continuous problem can be established independently as in \cite{irena}.
A systematic study of structure-preserving discretisations of the Ermakov-Painlev\'e II
equation is beyond the scope of this paper and is left for future investigation.

It is important to emphasize that the methods used for the continuous
Ermakov-Painlev\'e II model cannot, in general, be extended to its discretised version.
This is particularly true for the repulsive case, which requires the development of an
alternative method and leaves an open problem to be considered in a subsequent paper.
Our work thus bridges the gap between the well-established theory of continuous
Ermakov-Painlev\'e systems and the requirements of discrete analysis and computation.
By providing a rigorous treatment of this discrete model, we aim to open new avenues
for its application in numerical and physically discrete contexts.


\section{Preliminaries}\label{prelim}

In this section, we summarize some useful facts concerning second order discrete
equations that shall be employed throughout the paper.
Some short proofs are included for the reader's convenience.

As usual, we may identify the set of all the functions $u:\{0, \ldots, N\} \to \mathbb{R}$ with
$\mathbb{R}^{N+1}$ and denote $u(x):=u_x$.
The first and second order discrete derivatives of $u$ shall be denoted respectively by
\begin{gather*}
\Delta u_{x} := u_{x+1} - u_{x}\quad x=0,\ldots, N-1, \\
\Delta^2 u_{x-1} := \Delta u_{x} - \Delta u_{x-1} = u_{x+1} - 2u_{x} + u_{x-1}
\quad x=1,\ldots, N-1.
\end{gather*}
It is noted that the discretisation given by \eqref{disc-eq} follows the standard
procedure of a finite difference scheme for the continuous Ermakov-Painlev\'e II
equation \eqref{eq-cont}.
Indeed, letting $u_x:=y\left(\frac xN\right)$ for $x=0,\ldots, N$ and
taking into account that $y''\left(\frac xN\right)\simeq \frac 1{N^2} \Delta^2 u_{x-1}$,
the discrete equation \eqref{disc-eq} is obtained, with $a=\frac A{N^2}$,
$b=\frac B{N^3}$ and $c =\frac C{N^2}$.
For a more detailed exposition of this discretisation in a similar context,
see e.g.\ \cite{jan}.


Although $u$, $\Delta u$ and $\Delta^2u$ belong to different spaces, we shall
always employ the notation $\|\cdot\| _{2}$ to refer to the Euclidean norm, namely
\begin{gather*}
\|u\|_2 = \Big(\sum_{x=0}^N u_x^2\Big)^{1/2},\quad
\|\Delta u\|_2 = \Big(\sum_{x=0}^{N-1} (\Delta u_x)^2\Big)^{1/2}, \\
\|\Delta^2 u\|_2 = \Big(\sum_{x=0}^{N-2} (\Delta^2 u_x)^2\Big)^{1/2},
\end{gather*}

\begin{lemma}\label{parts}
Assume that $u_0=u_N=0$ or $\Delta u_0 = \Delta u_{N-1}=0$, then
$$
\|\Delta u\|_2^2 \le \|u\|_2 \|\Delta^2 u\|_2.
$$
\end{lemma}

\begin{proof}
We shall use the summation by parts formula, slightly adapted to the present context.
 A detailed deduction is included for the reader's convenience:
\begin{align*}
\sum_{x=1}^{N-1} \Delta^2u_{x-1}u_x
&=\sum_{x=1}^{N-1} (\Delta u_x - \Delta u_{x-1})u_x \\
&= \sum_{x=1}^{N-1} \Delta u_x u_x - \sum_{x=0}^{N-2} \Delta u_{x}u_{x+1} \\
&= \Delta u_{N-1} u_{N-1} - \Delta u_0 u_1
 +\sum_{x=1}^{N-2}\Delta u_{x}(u_{x}-u_{x+1}) \\
&= \Delta u_{N-1} u_{N-1} - \Delta u_0 u_1 - \sum_{x=1}^{N-2}(\Delta u_{x})^2.
\end{align*}
Under any of the two assumptions, it is immediately seen that
$$
\Delta u_{N-1} u_{N-1} - \Delta u_0 u_1= -(\Delta u_{N-1})^2 - (\Delta u_{0})^2,
$$
so the result follows from the Cauchy-Schwarz inequality.
\end{proof}

\begin{lemma}\label{eigen}
Assume that $u_0=u_N=0$ and let $\lambda_1:= 4\sin^2\left(\frac \pi{2N}\right)$, then
\begin{enumerate}
\item $ {\lambda_1} \|u\|^2_2\le \|\Delta u\|^2_2$;
\item $\|u\|_2\le \frac 1{\lambda_1}\|\Delta^2 u\|_2$.
\end{enumerate}
\end{lemma}

\begin{proof}
The first inequality follows from the fact that $\lambda_1$ is the first eigenvalue
of the problem
$$
-\Delta^2u_{x-1} =\lambda u_x\quad u_0=u_N=0,
$$
while the second one is deduced from the first one, combined with the preceding lemma.
\end{proof}

\begin{lemma}
Let $g_x:\mathbb{R}\to \mathbb{R}$ be continuous and bounded for $x=1,\ldots, N-1$.
Then the  equation
$$
\Delta^2u_{x-1} - u_x = g_{x}(u_x) \quad x=1,\ldots, N-1
$$
has at least one solution, both for arbitrary
Dirichlet conditions \eqref{dir} and Robin conditions
 \eqref{disc-robin} with $f_0, f_N:\mathbb{R}\to \mathbb{R}$ continuous and bounded.
\end{lemma}

\begin{proof}
For each $v\in \mathbb{R}^{N+1}$ and both conditions \eqref{dir} and \eqref{disc-robin},
define $T(v):=u$ as the unique solution of the linear problem
$$
Lu_x:=\Delta^2u_{x-1} - u_x = g_{x}(v_x) \quad x=1,\ldots, N-1
$$
satisfying \eqref{dir} or
$$
\Delta u_0 = f_0(v_0), \quad \Delta u_{N-1} = f_N( v_{N})
$$
respectively. It is verified that $T:\mathbb{R}^{N-1}\to \mathbb{R}^{N-1}$
is well defined and continuous. In fact, 
in both cases, the problem may be written in matrix form as
$\mathcal Au=\mathcal B$, where $\mathcal A$ is a (constant) invertible 
$(N-1)\times (N-1)$ tridiagonal matrix and $\mathcal B\in \mathbb{R}^{N-1}$ is a bounded vector
depending continuously on $v$. This is readily applied to prove that the range of the 
mapping $T$ is bounded, so by Brouwer's fixed point theorem the existence of at least 
one fixed point is deduced.
\end{proof}

The preceding lemma allows
to adapt to the present context the method of upper and lower solutions for a general equation
\begin{equation}
 \label{disc-eq-gral} \Delta^2 u_{x-1} = G_x(u_x)\quad x=1,\ldots, N-1
\end{equation}
with $G_x:\mathbb{R}\to \mathbb{R}$ continuous for $x=1,\ldots, N-1$.
We recall that the method was successfully extended to discrete second order problems under various boundary conditions, see e.g. \cite{AOR, AW, CO, FORP}. However, to the best of our knowledge, no results can be found in the literature for the nonlinear Robin condition \eqref{disc-robin}. For convenience, we shall assume without loss of generality
that the continuous functions $f_0$ and $f_N$, as well as $G_x$ for $x=1,\ldots, N-1$ are
defined on the whole line.

\begin{definition} \rm
We shall say that $\beta$ is an upper solution for \eqref{disc-eq-gral}-\eqref{dir} or
\eqref{disc-eq-gral}-\eqref{disc-robin} if
$$
\Delta^2 \beta_{x-1} \le G_x(\beta_x)\quad x=1,\ldots, N-1
$$
and
$$
\beta_0\ge D_0, \quad \beta_N\ge D_N
$$
or
$$
\Delta \beta_0 \le f_0(\beta_0),\quad \Delta \beta_{N-1} \ge f_N(\beta_N)
$$
respectively. A lower solution $\alpha$ is defined analogously, with all the 
inequalities reversed.
\end{definition}

\begin{lemma} \label{sub-sup} 
Assume there exist $\alpha$ and $\beta$ as before such that $\alpha_x\le \beta_x$ 
for all $x=0,\ldots, N$. Then the respective problem  \eqref{disc-eq-gral}-\eqref{dir} or
\eqref{disc-eq-gral}-\eqref{disc-robin}
has at least one solution $u$ such that $\alpha_x\le u_x\le \beta_x$ for all
 $x=0,\ldots, N$.
 \end{lemma}
 
\begin{proof}
Define as usual the  (continuous) truncation function
$$
\mathcal T_x(u):=
 \begin{cases}
 \beta_x & \text{if } u\ge \beta_x, \\
 u & \text{if } \alpha< u <\beta_x, \\
 \alpha_x & \text{if } u\le \alpha_x,
 \end{cases}
$$
and,  using the previous lemma,  set $u$ as a solution of the problem
 $$
\Delta^2 u_{x-1} - u_x = G_x(\mathcal T_x(u_x))-\mathcal T_x(u_x)\quad x=1,\ldots, N-1
 $$
under the condition \eqref{dir} or
 $$
\Delta u_0=f_0(\mathcal T_0(u_0)), \quad \Delta u_{N-1}=f_{N}(\mathcal T_N(u_{N}))
$$
respectively. Suppose for example that
$w:=u-\beta$ achieves a maximum at some
$x =1,\ldots, N-1$, then 
$$
\Delta^2 w_{x-1}= w_{x+1}-w_x + w_{x-1}-w_x\le 0,
$$
that is, $\Delta^2 u_{x-1} \le \Delta^2 \beta_{x-1}$. If $u_x>\beta_x$, then
$$
\Delta^2 u_{x-1} > G_x(\beta_x) \ge \Delta^2 \beta_{x-1},
$$
a contradiction. In the same way, it is verified that $u-\alpha$ cannot achieve 
a negative minimum at $x=1,\ldots, N-1$. In the Dirichlet case, it is also clear that 
$\alpha_x\le u_x\le \beta_x$ for $x=0$ and $x=N$; thus, $u$ lies between $\alpha$ and 
$\beta$ for all $x$ and solves the original problem. For the Robin conditions, 
suppose that $u-\beta$ does not achieve a maximum at $x=1,\ldots, N-1$, then a strict 
absolute maximum is achieved either at $x=0$ or $x=N$. In the first case, if moreover
$u_0 >\beta_0$, then $u_0-\beta_0 > u_1-\beta_1$, that is
$$
\Delta \beta_0 \le f(\beta_0) = \Delta u_0 < \Delta \beta_0,
$$
a contradiction. The proof is analogous for
$x=N$, so we are able to conclude that $\beta\ge u$. Similarly, it is
verified that $\alpha\le u$, whence $u$ is a solution of the original problem.
 \end{proof}
 
\section{Attractive case}

Throughout this section, we shall assume that the singularity is attractive, that is,
$c<0$.

\subsection{General results}
Applying the lemmas of the previous section, we shall establish our results. 
Recall that the parameters $a,b,c$ depend on $N$ as defined in Section \ref{prelim},
 a fact that will have significance in some of the conditions described below. 

\begin{theorem}\label{attract-1}
Let $a> 0 >c$. Then \eqref{disc-eq}-\eqref{dir} admits at least one positive solution 
for arbitrary $D_0,D_N>0$ and \eqref{disc-eq}-\eqref{disc-robin} admits at least one positive solution, 
provided that
$$
f_0(\alpha), f_N(\beta) \le 0\le f_N(\alpha), f_0(\beta)
$$
for some $\alpha >0$ small enough and some $\beta >\alpha$ large enough.
Furthermore, let 
$$
M:= -\frac 9{N-1}\Big(\frac{a^2c}4\Big)^{1/3} > 0
$$
and set
$$
\hat b:= \begin{cases}
 \frac b{N-1} & \text{if } b\ge 0 \\
 b & \text{otherwise}.
\end{cases} 
$$
Then there are no other positive solutions if
 \begin{equation}\label{uniq}
 \hat b + M > - \frac 4{N-1} \sin^2\Big(\frac \pi{2N}\Big)
 \end{equation}
for the Dirichlet condition, and
\begin{equation}\label{uniq-rob}
\hat b + M > 0
\end{equation}
in the Robin case, provided that $f_0$ is nondecreasing and $f_N$ is nonincreasing.
\end{theorem}

\begin{proof}
Define $G_x(t):= a t^3 + bxt + \frac c{t^3}$, 
then taking constants $\alpha>0$ sufficiently small and $\beta>0$ sufficiently large
it is seen that $G_x(\alpha) < 0 < G_x(\beta)$ for all $x=0,\ldots, N$.
In particular, $\alpha$ and $\beta$ may be chosen in such a way that all the conditions 
of Lemma \ref{sub-sup} are satisfied, so the existence of at least one solution follows.
Now assume that $u$ and $v$ are positive solutions and set $w:=u-v$. 
Firstly observe that, if $s,t>0$ then
$$
\Big[a(s^3-t^3) + c\Big(\frac 1{s^3} - \frac 1{t^3} \Big)\Big] (s-t) 
= \psi(s,t)(s-t)^2,
$$
where $\psi(s,t):=\left( a - \frac c{s^3t^3}\right)(s^2+st+t^2)$ and, by computing 
the critical points of $\psi$, it is readily seen that 
$\psi(s,t)\ge -9\big(\frac{a^2c}4\big)^{1/3}$ for all $s,t>0$. Thus, 
$$
\sum_{x=1}^{N-1} \Delta^2 w_{x-1}w_{x}
= \sum_{x=1}^{N-1} [G_x(u_x)-G_x(v_x)](u_x-v_x)\ge (N-1)M\sum_{x=1}^{N-1} w_x^2 +
\sum_{x=1}^{N-1}xbw_x^2,
$$
whence
\begin{align*}
&-\|\Delta w\|_2^2
+ w_N\Delta w_{N-1} - w_0\Delta w_0 \ge
(N-1)M\sum_{x=1}^{N-1} w_x^2 + \sum_{x=1}^{N-1}xbw_x^2 \\
& \ge (N-1)(M+\hat b)\sum_{x=1}^{N-1} w_x^2, \quad \text{since $1\le x\le N-1$.}
\end{align*}

On the one hand, in the Dirichlet case, it is seen that $w_0=w_N=0$, so the previous 
inequality, combined with the assumption \eqref{uniq} yields
%textcolor{blue}
$$
-\|\Delta w\|_2^2 \ge (N-1)(M+ \hat b)\|w\|_2^2 > -\lambda_1\|w\|_2^2,
$$
and the proof follows from Lemma \ref{eigen}.

On the other hand, under the Robin
condition, the monotonicity assumptions on $f_0$ and $f_N$ imply
$$
w_N\Delta w_{N-1} \le 0 \le w_0\Delta w_0;
$$
thus, the assumption $M + \hat b>0$ directly implies
that
$$
\|\Delta w\|^2 = \sum_{x=1}^{N-1} w_x^2=0.
$$
We conclude that $w$ is constant and $w_x=0$ for $x=1,\ldots, N-1$ which, in turn, 
implies $w_0=w_N=0$.
\end{proof}

\begin{remark}\label{conv-cont} \rm
It is interesting to compare the previous result with the corresponding one 
for the continuous equation \eqref{eq-cont}; particularly, the uniqueness conditions
\eqref{uniq} and \eqref{uniq-rob} which, letting $N\to \infty$, yield
$$
 \min \{B,0\} - 9\Big(\frac{A^2C}4\Big)^{1/3} \ge -\pi^2
$$
and $B\ge 0$  respectively. 
The latter two inequalities can be easily obtained from \eqref{eq-cont} by direct
computation and, in the Dirichlet case, the condition given in \cite{pacr15} is thus 
improved.
\end{remark}

Next, we shall consider the case $a\le 0$. In the previous paper \cite{pacr15}, 
a somewhat sharp sufficient condition was obtained by considering the solutions of 
the autonomous problem
$$
u''(x)= au(x)^3 - b^-u(x) + \frac c{u(x)^3}, \quad 0<x<1
$$
as upper solutions of the original continuous version of \eqref{disc-eq}.
However, the bounds given in \cite{pacr15} strongly
relied on the fact that the autonomous equation can
be integrated when multiplied by $u'(x)$,  a procedure
that cannot be imitated here, due to the failure of the chain rule.
In contrast with that, the discrete problem has an advantage with respect to the 
continuous one, for which a constant upper solution $\beta>0$ does not exist when 
$a\le 0$, since the inequality
$$
bx\beta\ge -\Big(a\beta^3 + \frac c{\beta^3} \Big)>0 
$$
needs to be satisfied for all $x\in (0,1)$.
As a consequence, the following result provides sufficient conditions for the 
existence of a constant upper solution for \eqref{disc-eq}
that depend on $N$ and do not survive in the limit $N \to \infty$ when $a<0$.

\begin{theorem}\label{th2}
Assume $c<0<b$, then:
\begin{enumerate}
 \item For $a=0$, the Dirichlet problem \eqref{disc-eq}-\eqref{dir}
 has a unique positive solution. If furthermore
$$
 f_0(\alpha) \le 0\le f_N(\alpha),\quad
 f_0(\beta)\ge 0\ge f_N(\beta)
$$
for some sufficiently small $\alpha>0$ and some sufficiently large 
$\beta>\alpha$, then the nonlinear Robin problem \eqref{disc-eq}-\eqref{disc-robin} 
has also a positive solution, which is unique when $f_0$ is nondecreasing and 
$f_N$ is nonincreasing.

\item For $a<0$, assume that
 \begin{equation}\label{beta-cond}
 4b^3 \ge -27ca^2,
 \end{equation}
and define
\begin{equation}\label{beta}
\beta = \sqrt{\frac{-2b}{3a}}.
\end{equation}
Then \eqref{disc-eq} has at least one positive solution satisfying \eqref{dir} 
or \eqref{disc-robin}, provided that
$$
 \beta \ge D_0, D_N
$$
and
$$
f_0(\alpha) \le 0\le f_N(\alpha),\quad f_0(\beta)\ge 0\ge f_N( \beta)
$$
for some sufficiently small $\alpha>0$,
respectively.
\end{enumerate}
\end{theorem}

\begin{proof} 
The proof for the case $a=0$ follows exactly as in the preceding result. For $a<0$, let $G_x$ be defined as in the previous proof, then clearly $G_x\ge G_1$, so it suffices to verify that $G_1(%
{\beta})\ge 0$. In turn, this is equivalent to inequality \eqref{beta-cond},
 so the result follows.
\end{proof}

 \begin{remark} \rm
According to the preceding comment regarding the continuous model 
\eqref{eq-cont}, it is observed, in the second case, that \eqref{beta-cond} cannot 
be satisfied when $N$ is large.
\end{remark}

It is observed, in the latter proof, that  the constant $\beta=\beta(b)$
is computed as the (unique) positive value in which the absolute maximum of the
function $\varphi(z):= az^6 + bz^4$ with $z\ge 0$ is achieved. 
This shows that, if the inequality in \eqref{beta-cond} is strict, then the conditions 
may be relaxed by considering the maximal interval $\mathcal I_c\subset (0,+\infty)$ 
such that $\varphi(z)\ge -c$ for $z\in I_c$.
With this in mind, let us set the notation $M_c$ for the upper endpoint of 
$\mathcal I_c$.

\begin{corollary}
 For $a,c < 0 < b$, assume that \eqref{beta-cond} holds and let
$\mathcal I_c$ and $M_c$ be defined as before. Then
\begin{enumerate}
 \item Problem \eqref{disc-eq}-\eqref{dir} has at least one positive solution for 
 arbitrary $D_0, D_N\in (0,M_c]$.
 
 \item Problem \eqref{disc-eq}-\eqref{disc-robin} has at least one positive solution, 
 provided that
 $$
 f_0(\alpha) \le 0\le f_N(\alpha),\quad
 f_0(\beta)\ge 0\ge f_N(\beta)
 $$
for some sufficiently small $\alpha>0$ and some $\beta\in \mathcal I_c$.
\end{enumerate}
\end{corollary}

The next corollary is directly deduced from the fact that,
as $b\to +\infty$, the value
$\beta(b)$ tends to $+\infty$ and the lower endpoint of $\mathcal I_c$ tends to $0$.

\begin{corollary} 
Assume that $a, c< 0$, then:
\begin{enumerate}
 \item Given arbitrary $D_0, D_N>0$, there exists $b^*>0$ such that problem 
 \eqref{disc-eq}-\eqref{dir} has at least one positive solution if $b\ge b^*$.
 \item There exists $b^*>0$ such that problem \eqref{disc-eq}-\eqref{dir} has at 
 least one positive solution if $b\ge b^*$, provided that
$$
 f_0(\beta)\ge 0\ge f_N(\beta)
$$
for some $\beta >0$ and
$$
f_0(\alpha_n) \le 0\le f_N(\alpha_n)
$$
for some sequence $\alpha_n\to 0^+$.
\end{enumerate}
\end{corollary}

In the same spirit of the latter corollary, it is seen that solutions exist when the 
negative constant $c$ is close enough to $0$.

\begin{corollary} 
Assume that $a< 0 < b$ and define 
 $\beta=\beta(b)$ as before. Then:
\begin{enumerate}
\item There exists $c_*>0$ such that \eqref{disc-eq}-\eqref{dir} has at least 
 one positive solution for $-c_*<c<0$, provided that  $\beta\ge D_0, D_N$.

\item There exists $c_*>0$ such that \eqref{disc-eq}-\eqref{disc-robin} has at least 
one positive solution for $-c_*<c<0$, provided that
 \begin{enumerate}
 \item $f_0({\beta}) \ge 0 \ge f_N({\beta})$.
 \item $f_0(\alpha_n)\le 0\le f_N(\alpha_n)$ for some sequence $\alpha_n\to 0^+$.
\end{enumerate}
\end{enumerate}
\end{corollary}

\subsection{Homogeneous Dirichlet problem}

In this section, we shall consider problem \eqref{disc-eq} under the homogeneous 
condition
\begin{equation}
 \label{dir0} u_0=u_N =0.
\end{equation}
In contrast with the continuous case,
here the singularity at the boundary does not imply that the derivatives are unbounded. 
In fact, if a sequence of solutions is bounded, then so is the sequence of its 
derivatives. This suggests us to look for a solution of \eqref{disc-eq}-\eqref{dir0} as 
the limit of a sequence of positive solutions of \eqref{disc-eq}-\eqref{dir}
with Dirichlet conditions converging to $0$.
Such a procedure yields the following result.

\begin{theorem} \label{attract-hom} 
Problem \eqref{disc-eq}-\eqref{dir0} has at least one solution with $u_x >0$ for 
$x=1,\ldots, N-1$, provided that one of the following conditions holds:
 \begin{enumerate}
 \item $a > 0>c$.
 \item $a=0$ and $b>0>c$.
 \item $a,c<0$ and $4b^3 > -27ca^2$.
 \end{enumerate}
Furthermore, the solution is unique in the second case, and also in the first case 
under assumption \eqref{uniq}.
\end{theorem}

\begin{proof} 
Fix an arbitrary sequence $r_k\searrow 0$ and a constant $\beta>0$ such
that $G_x(\beta)\ge 0$ for $x=1,\ldots, N-1$.
According to Lemma \ref{sub-sup}, $\beta$
serves as an upper solution of the Dirichlet problem when
$D_0, D_N\le \beta$.
Thus, for each $k\in\mathbb N$ we may assume that $r_k$ is a lower solution
with $r_k< \beta$ for all $k$
and define {$u^{(k)}$} as a solution of \eqref{disc-eq} between $r_k$ and 
$\beta$ satisfying $u^{(k)}_0=u^{(k)}_N = r_k$.
Since $u^{(k)}_x\le \beta$ for all $x=0,\ldots,N$, taking a subsequence we may assume 
that $\{u^{(k)}\}$ converges to some $u$ such that $u_0=u_N=0$. Furthermore, the 
identity
$$
 \Delta^2 u^{(k)}_{x-1} = a (u^{(k)}_{x})^3 + bxu^{(k)}_{x} 
 + \frac c{(u^{(k)}_x)^3} \quad x=1,\ldots, N-1,
$$
combined with the fact that the left-hand side term is bounded imply that $u_x>0$ and
$u$ verifies \eqref{disc-eq} for $x=1,\ldots, N-1$.
Uniqueness is deduced exactly as before.
\end{proof}

 \begin{remark} \rm
Again, condition (3) in the previous theorem depends on $N$ and, not surprisingly,
for fixed $A,C<0$, it does not hold when $N$ is large.
 \end{remark}

\section{Repulsive case}

Here, we deal with the case $c>0$.
We shall consider firstly the Dirichlet
problem.

\begin{theorem}
Assume $c>0>a$, then
\eqref{disc-eq}-\eqref{dir} with $D_0, D_N\ge 0$ has at least one solution $u$ with 
$u_x>0$ for $x=1,\ldots, N-1$.
\end{theorem}

\begin{proof}
Let us firstly observe that the problem may be also written as
$$
 u_{x+1} + u_{x-1} = a u_{x}^3 + (2+bx)u_{x} + \frac c{u_x^3} \quad x=1,\ldots, N-1,
$$
where it is already assumed that $u_0=D_0$ and $u_N=D_N$.
This shows that $u$ is a (positive) solution if and only if $P_x(u)=0$ for all 
$x=1,\ldots, N-1$, where
$$
P_x(u)= au_x^6 + (2+bx)u_x^4 -(u_{x-1} + u_{x+1}) u_x^3 + c.
$$
Set
$$
\Omega:=\{ u\in \mathbb{R}^{N-1}: 0<u_x< R, \; x=1,\ldots, N-1 \}
$$
for some $R>0$ to be defined, and the
homotopy $H:\overline\Omega\times [0,1]\to \mathbb{R}^{N-1}$ given by
$$
H(u,\lambda)_x:=
au_x^6 + \lambda \left[(2+bx)u_x^4 -(u_{x-1} + u_{x+1}) u_x^3\right] + c.
$$
It is observed that, if $u_x=0$ for some $x$, then 
$H_\lambda (u)_x:=H(u,\lambda)_x=c\ne 0$. Furthermore, taking $R$ sufficiently large 
it is also clear that also
$H_\lambda (u)_x\ne 0$ when $u_x=R$; in other words, we have verified that
$H_\lambda$ does not vanish on $\partial \Omega$ for arbitrary $\lambda\in [0,1]$. 
From the homotopy invariance of the Brouwer degree, we deduce that
$$
\deg(H_1,\Omega,0)= \deg(H_0,\Omega,0)= (-1)^{N-1},
$$
where the latter equality follows from the
fact that $H_0(u)_x=au_x^6+c$, which has a
unique simple positive root $u_*:= \sqrt[6]{\frac{-c}a}$ and the sign of the 
Jacobian determinant of $H_0$ at $(u_*,\ldots, u_*)$ coincides with that of $a^{N-1}$.
Hence, the existence property of the Brouwer degree implies that
$H_1$ has at least one root in $\Omega$,
which corresponds to a solution of the problem.
\end{proof}

\begin{remark} \label{polynomials} \rm
\begin{enumerate}
 \item The previous proof is also valid when $c<0<a$; thus, the existence part of
 Theorem \ref{attract-1} is retrieved, including the homogeneous case treated in 
 Theorem \ref{attract-hom}.
 In both situations, the fact that $ac<0$ allows the direct use the 
 Poincar\'e-Miranda theorem instead of the Brouwer degree.
 Indeed, it suffices to observe, for $u\in \overline \Omega$, that
$$
 P_x(u)=c\quad \text{if $u_x=0$}
$$
and
$$
P_x(u)\sim aR^6\quad \text{if $u_x=R$},
$$
which shows that $P_x$ changes sign at the corresponding faces of $\Omega$.
However, the topological degree is more general and may constitute an useful tool 
when searching for multiple solutions.

\item It is worth recalling,
for the non-homogeneous continuous case \eqref{eq-cont}, that the assumption 
$A<0<C$ yields in fact the existence of infinitely many solutions. 
More precisely, for all $k\in \mathbb N$ sufficiently large there exist at least two 
solutions with exactly $k$ nodal regions with respect to the segment joining the 
boundary values.
Such a result cannot be reproduced in the discrete model for obvious reasons, 
although it is expected that the number of solutions increases as $N$ gets larger.
As a toy model, let $N=2$,
then the problem reduces to find the zeros of the polynomial
$$
P_1(u_1)=au_1^6 + (2+b)u_1^4 -(D_{0} + D_{2}) u_1^3 + c,
$$
which trivially verifies
$$
P_1(0)= c>0,\quad P_1(u_1)\to -\infty\quad\text{as $u_1\to+\infty$}.
$$
By computing its derivative (or simply by the Descartes rule), it is easy to 
conclude that there exist at most $3$ positive solutions; moreover,
the solution is unique if and only if
$$
b+2 < \frac 94 \Big(\frac {-a(D_{0} + D_{2})^2}2 \Big)^{1/3}.
$$
\end{enumerate}
\end{remark}

Next, we proceed with the Robin case.

\begin{theorem}\label{rob-rep}
Assume $a<0<c$ and
\begin{enumerate}
\item $f_0(\eta) \le 0 \le f_N(\eta)$ for some $\eta>0$ small enough.

\item There exists a sequence $R_n\to+\infty$ such that
$$
 \liminf_{n\to\infty} \frac {f_0(R_n)}{R_n} > -1, \quad \limsup_{n\to\infty} \frac {f_N(R_n)}{R_n} < 1.
$$
\end{enumerate}
Then \eqref{disc-eq}-\eqref{disc-robin} has at least one positive solution.
\end{theorem}

\begin{proof}
We shall follow the outline of the previous proof, now taking into account the Robin 
condition, which motivates to define
$$
P_0(u):=f_0(u_0) + u_0 - u_1,\quad P_N(u):=-f_N(u_N) + u_N - u_{N-1}.
$$
Within this context, the polynomials $P_x$ for $x=1,\ldots x_{N-1}$ are defined as 
before for arbitrary $u\in (0,+\infty)^{N+1}$.
It is {noted} that the homotopy of the preceding proof may be extended by setting
$$
H_\lambda(u)_0:= \lambda f_0(u_0) + u_0 - u_1,\quad
H_\lambda(u)_N:= -\lambda f_N(u_N) + u_N - u_{N-1}.
$$
According to the assumptions, we may
fix $\varepsilon>0$ and $n_0$ such that
$$
f_0(R_n) + R_n > \varepsilon R_n, \quad f_N(R_n) - R_n < -\varepsilon R_n
$$
for all $n\ge n_0$.
Next, fix $R:= R_n$ for some sufficiently large $n\ge n_0$ such that if 
$H_\lambda (u)_x =0$ for all $x=1,\ldots, N-1$ with
$0< u_y \le R$ for all $y=0,\ldots, N$, then $u_x \le \varepsilon R$ for 
$x=1,\ldots, N-1$.
Suppose now that $H_\lambda(u) = 0$ with $0<u_x\le R$ for all $x$ and $u_0=R$, then
$$
\lambda f_0(R) + R = u_1\le \varepsilon R.
$$
This implies $f_0(R)<0$ and, consequently,
$$
\lambda f_0(R) + R \ge f_0(R) + R >\varepsilon R,
$$
a contradiction. In the same way, it is verified that,
if $H_\lambda(u) = 0$ with $0<u_x\le R$, 
then $u_N<R$. Now fix  $\eta>0$ such that
if $0<u_x\le R$ for all $x=0,\ldots N$ and
$u_j\le\eta$ for some $j=1,\ldots, N-1$, then
$H_\lambda(u)_j\ne 0$. Suppose that $H_\lambda(u)=0$ with $\eta \le u_x<R$ for all 
$x$ and $u_0=\eta$, then
$$
 \lambda f_0(\eta) + \eta =u_1 >\eta.
$$
From the hypothesis, $\eta$ may be chosen in such a way that $f_0(\eta)\le 0$, 
which yields a contradiction. In the same way, we deduce that $u_N>\eta$, whence 
the homotopy does not vanish on $\partial\Omega$, where
$$
\Omega:= \{ u\in \mathbb{R}^{N+1}: \eta<u_x<R,\quad x=0,\ldots, N\}.
$$
Again, this implies
$$
\deg(H_1,\Omega,0)=\deg(H_0,\Omega,0)=(-1)^{N+1}
$$
because the Jacobian matrix of $H_0$ at the unique root $(u_*,\ldots,u_*)$ is now 
given by
$$
\begin{pmatrix}
 1 & -1 & 0 & \ldots & 0 \\
 0 & 6au_*^5 & 0 &\ldots & 0\\
 0 & \ldots &\ldots & \ldots & 0\\
 0 & 0 &\ldots & 6au_*^5 & 0\\
 0 & 0 & \ldots & -1 & 1
 \end{pmatrix}.
 $$
 \end{proof}

Next, we shall focus on the case $a,c>0$.
In the Dirichlet case, it is immediately seen that solutions cannot exist when $c$ 
is sufficiently large.
The same happens in the Robin case, provided that $f_0$ and $f_N$ satisfy some 
appropriate growth conditions.
The following result shows that solutions exist in the opposite situation.

\begin{theorem}\label{a, c>0}
Let $a>0$, then:
\begin{enumerate}
 \item There exists $c_*>0$ such that problem \eqref{disc-eq}-\eqref{dir} has at 
 least one positive solution for $0<c<c_*$, in the following cases:
\begin{enumerate}
 \item $D_0>0$ or $D_N>0$.
 \item $D_0=D_N=0$ and $b < -\frac 2{N-1}$.
\end{enumerate}

\item There exists $c_*>0$ such that problem \eqref{disc-eq}-\eqref{disc-robin} has 
at least one positive solution for $0<c<c_*$, provided that
$$ 
 {f_0(r_0)} + r_0 \le 0 \le {f_N(r_N)} - r_N
$$
for some $r_0, r_N>0$ and 
$$
\liminf_{n\to\infty} \frac {f_0(R_n)}{R_n} > -1, \quad 
\limsup_{n\to\infty} \frac {f_N(R_n)}{R_n} < 1
$$
 for some sequence $R_n\to+\infty$.
\end{enumerate}
\end{theorem}

\begin{proof} 
Consider firstly the Dirichlet case with $D_0>0$ and define the function
$$
Q_x(u):= P_x(u) - c = u_x^3[au_x^3 + (2+bx)u_x -(u_{x-1} + u_{x+1})].
$$
Next, fix $\varepsilon_1>0$ such that
$$
at^3 + (2+b)t < D_0\quad 0<t\le \varepsilon_1
$$
and, inductively, we may set $\varepsilon_2,\ldots, \varepsilon_{N-1}>0$
such that
$$
at^3 + (2+bx)t < \varepsilon_{x-1}\quad 0<t\le \varepsilon_{x}.
$$
Also, we may fix $R>0$ such that $aR^3 + (2+bx)R > 2R$ for $x=1,\ldots, N-1$ and 
define the open set
$$
\Omega:= (\varepsilon_1,R)\times\ldots\times (\varepsilon_{N-1},R).
$$
It follows from the previous choice of the values $\varepsilon_x$ and $R$
that if $u\in\partial \Omega$, then there exists $x$ such that either
$u_x=R$ and $Q_x(u)>0$, or $u_x=\varepsilon_x$ and $Q_x(u)<0$. 
Thus, picking an arbitrary point $v\in \Omega$, in both situations it is deduced that
$\lambda Q_x(u) + (1-\lambda)(u_x - v_x)\ne 0$ for $\lambda\in [0,1]$. 
In other words, the homotopy
$H_\lambda:= \lambda Q +(1-\lambda)(I-v)$ does not vanish, where $I$ denotes the identity 
map and, in consequence,
$$
\deg(Q,\Omega,0)= \deg(I-v,\Omega,0)=1.
$$
Because the degree is continuous, we conclude that
$$
\deg(P ,\Omega,0)= \deg(Q+ (c,\ldots,c), \Omega,0)=1
$$
when $c$ is sufficiently small.
The proof is analogous when $D_0=0<D_N$, now starting from an appropriate value 
$\varepsilon_{N-1}>0$ such that
$$
at^3 + (2+b(N-1))t < D_N\quad 0<t\le \varepsilon_{N-1}
$$
and, inductively, choosing values $\varepsilon_{N-2},\ldots, \varepsilon_1>0$ such that
$$
at^3 + (2+bx)t < \varepsilon_{x+1}\quad 0<t\le \varepsilon_{x}.
$$
Finally, if $D_0=D_N=0$ and $b < -\frac 2{N-1}$, it suffices to fix $\varepsilon_{N-1}>0$ 
such that
$$
at^3 + (2+b(N-1))t < 0\quad 0<t\le \varepsilon_{N-1}
$$
and proceed as before.

Concerning the Robin case, set $\varepsilon_0:=r_0$, $\varepsilon_N:=r_N$ and 
$\varepsilon_x$ as before for $x=1,\ldots, N-1$. Then, fix a value $R>0$ as in the proof 
of Theorem \ref{rob-rep} and set
\begin{gather*}
\tilde\Omega:= (\varepsilon_0,R)\times\ldots\times (\varepsilon_{N},R), \\
\tilde Q(u);= (f_0(u_0)+ u_0-u_1, Q(u_1,\ldots,u_{N-1}), -f_N(u_N)+ u_N-u_{N-1}).
\end{gather*}
Again, it is deduced that $\deg(\tilde Q, \tilde \Omega,0)=1$, and the proof 
follows from the fact that $P = \tilde Q + (0,c,\ldots, c, 0)$.
\end{proof}

As a conclusion of this section, let us emphasize that the  case $a<0<c$ can be
analysed by a more careful study of the associated polynomial system and, as
mentioned in Remark \ref{polynomials}, the non-homogeneous Dirichlet problem 
for \eqref{eq-cont} has infinitely many solutions.
In contrast, the discretization involves a system of $N-1$ polynomials with $N-1$ 
variables and, as is well known, the number of (nondegenerate) positive solutions is 
always finite. A celebrated -although unsharp- upper bound of this number was given 
by Khovanskii in \cite{kho} and improved in later works; in our case, the system can 
be reduced to a single equation, so an upper bound is readily obtained from Descartes' 
rule of signs.
However, no general results for lower bounds seem to exist in the literature.
Such bounds would be of interest, since it is expected that the number of solutions 
tends to infinity as $N\to\infty$.
Concerning the case $a,c>0$, existence of solutions of \eqref{disc-eq}-\eqref{dir} 
is guaranteed when the associated polynomial system has at least one positive root but, 
again, the literature on this topic is scarce.
Some recent results have been obtained (see e.g.\ \cite{wang}),
but the conditions do not apply to our specific case.
Concerning the case $a,c>0$, we emphasize that the threshold $c_*$ in
Theorem \ref{a, c>0} depends on $N$, so the result is therefore of a 
purely discrete nature. In particular, it is not intended to be carried over to the 
continuous limit $N\to\infty$.

 \section{Variational approach}

In this section, we shall briefly describe the main aspects of the variational method, 
adapted to the present boundary value problems.
This aims, on the one hand, to give alternative proofs of some of the preceding results 
and, on the other hand, to propose a useful tool in order to tackle the multiplicity 
problem suggested in the previous section, with the help of linking-type theorems.
This goal shall be pursued in a forthcoming paper.

For the Dirichlet case, define the functional
$\mathcal I:(0,+\infty)^{N-1}\to \mathbb{R}$ given by
\begin{equation}
\label{functional}
\mathcal I(u)= \frac 12\sum_{x=1}^N (\Delta u_{x-1})^2+\frac a4 \sum_{x=1}^{N-1} u_x^4
+ \frac b2 \sum_{x=1}^{N-1} xu_x^2 -
 \frac c2 \sum_{x=1}^{N-1} \frac 1{u_x^2},
\end{equation}
where it is assumed that $u_0=D_0$ and $u_N=D_N$.
 It is straightforwardly verified that the critical points of $\mathcal I$ coincide 
with the positive solutions of \eqref{disc-eq}-\eqref{dir}. 
Next, for $x=1,\ldots, N-1$ compute
$$
\frac {\partial \mathcal I}{\partial u_x}= 2u_x - (u_{x-1} + u_{x+1})
+ au_x^3 + bx u_x + \frac c{u_x^3}.
$$
This shows that $u$ is a critical point of $\mathcal I$ if and only if $P_x(u)=0$ for 
all $x=1,\ldots, N-1$, where $P_x$ are the polynomials defined in the previous section.

Analogously for condition \eqref{disc-robin},
the functional $\mathcal J:(0,+\infty)^{N+1}\to \mathbb{R}$ defined by
$$
\mathcal J(u):= \mathcal I(u) + F_0(u_0) - F_N(u_N),
$$
with $\mathcal I$ defined as in \eqref{functional} and
$$
F_0(s):=\int_0^s f_0(t)\, dt, \quad F_N(s):=\int_0^s f_N(t)\, dt.
$$
Here, if $u$ is a critical point of
 $\mathcal J$, then taking all the directional derivatives 
$\frac{\partial \mathcal J}{\partial \nu}(u)$
with $\nu_0=\nu_N=0$ it is deduced
that $u$ satisfies \eqref{disc-eq}.
Next, for arbitrary $\nu\ne 0$, summation by parts yields
$$0= \frac{\partial \mathcal J}{\partial \nu}(u) =
f_0(u_0)\nu_0 - f_N(u_N)\nu_N +\Delta u_{N-1}\nu_N -\Delta u_0\nu_0,
$$
so the Robin condition is verified by
taking firstly $\nu_0=0\ne \nu_N$ and next $\nu_0\ne 0= \nu_N$.
In order to verify that the critical
points of $\mathcal J$ coincide with the solutions of the problem, 
let us simply observe that
\begin{gather*}
\frac{\partial \mathcal J}{\partial u_x} 
= \frac{\partial \mathcal I}{\partial u_x} \quad x=1,\ldots, N-1, \\
\frac{\partial \mathcal J}{\partial u_0}=
f_0(u_0) + u_0 - u_1,\quad \frac{\partial \mathcal J}{\partial u_N} 
= -f_N(u_N) + u_N - u_{N-1}.
\end{gather*}
It is observed that the variational formulation provides an immediate resolution of 
the cases in which $a$ and $c$ have different signs, since a coercivity argument can be 
applied. For the Dirichlet case
assume, for example, that $a>0>c$, then $\mathcal I(u)\to+\infty$ as some coordinate of
$u$ tends to $0$ or to $+\infty$ which, in turn, implies that $\mathcal I$ achieves an
absolute minimum on $(0,+\infty)^{N-1}$. A similar conclusion follows when $a<0<c$ and 
also for the Robin case, under appropriate assumptions on $f_0$ and $f_N$.


\section{Discussion and concluding remarks}

A discretisation of the Ermakov-Painlev\'e II equation subject to certain boundary 
conditions has been carried out using a standard finite difference scheme, and both 
topological and variational approaches have been applied to analyse the resulting 
discrete problem.

Although a full numerical study is beyond the scope of this paper, we note that the 
convergence of solutions of the discrete scheme \eqref{disc-eq} to solutions of the 
continuous problem \eqref{eq-cont} can be established following the approach in 
\cite{irena} for singular mixed boundary value problems.
Moreover, as shown in Remark \ref{conv-cont}, the condition obtained here for 
uniqueness in the attractive case
improves those previously known for the continuous equation \eqref{eq-cont}
in the Dirichlet setting, and they reduce to the standard
one in the linear Robin context. 

The extension of the variational approach to a wider range of discretisation problems 
(e.g., other singular nonlinearities or boundary conditions) is the subject of 
current investigation.

\subsection*{Acknowledgements}

The authors thank the anonymous reviewers for the careful reading of the manuscript 
and the detailed list of suggestions and comments.
P. Amster was partially supported by projects
 UBACyT 20020190100039BA and PIP 11220200100175CO, CONICET.



\end{document} 